



Azienda
Ospedaliero
Universitaria
Careggi

SOD MEDICINA INTERNA E POST-CHIRURGICA

Direttore Prof. Carlo Rostagno

Largo Brambilla 3, 50134 Firenze FI
Tel. 055.794.8063 -8085 fax 055.794.8263 / 055.422406



UNIVERSITÀ
DEGLI STUDI
FIRENZE
Scuola di
Scienze della
Salute Umana

To the Editors in Chief of
World Journal of Cardiology

Dear Sir

I would like to submit to World Journal of Cardiology the revised manuscript “ Heart valve disease in elderly “ **Number ID:** 02729101.

We thank the reviewer for their appreciation . The English form was revised by a native English speaker .
One figure was added to the text . References were modified according to guidelines

Firenze January 13 2018

Carlo Rostagno

Corresponding author : Carlo Rostagno MD , PhD , Associate Professor of Internal Medicine
University of Florence , Chief of Internal Medicine and post-surgery Unit , AOU Careggi Florence
E mail carlo.rostagno@unifi.it , Address largo Brambilla 3 50134 Firenze tel +390557948545